Author’s response to reviews

Title: Protocol for the BAG-RECALL Clinical Trial: A Prospective, Multi-Center, Randomized, Controlled Trial to Determine Whether a Bispectral Index-Guided Protocol is Superior to an Anesthesia Gas-Guided Protocol in Reducing Intraoperative Awareness with Explicit Recall in High Risk Surgical Patients

Authors:

Michael S Avidan (avidanm@wustl.edu)
Ben J Palanca (palancab@wustl.edu)
David Glick (DGlick@dacc.uchicago.edu)
Eric Jacobsohn (EJacobsohn@exchange.hsc.mb.ca)
Alex Villafranca (ajv1982@gmail.com)
Michael O’Connor (moc5@dacc.uchicago.edu)
George A Mashour (gmashour@med.umich.edu)

Version: 2 Date: 11 November 2009

Author’s response to reviews:

Additional files - You have uploaded your ethics and funding documents as additional files. As these are confidential documents, please remove these from the manuscript file.
DONE.

Competing interests - Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.
DONE.

The questions that are asked of authors are:

Financial competing interests

- In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

- Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

- Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.
- Do you have any other financial competing interests? If so, please specify.
Non-financial competing interests
Are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.
THERE ARE NONE. THIS IS INCLUDED. THE ANSWERS TO ALL THE ABOVE QUESTIONS IS NO.

Authors’ contributions - Please include an Authors’ contributions section before the Acknowledgements and Reference list.
DONE.

For the Authors’ contributions we suggest the following kind of format (please use initials to refer to each author’s contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.
DONE.